Author's response to reviews

Title: Ethnic disparity in 21-hydroxylase gene mutations identified in Pakistani congenital adrenal hyperplasia patients

Authors:

Aysha Habib (aysha.habib@aku.edu)
Muniba Aban (muniba.aban@aku.edu)
Jamal Raza (jamalraza@yahoo.com)
Naeem ul Haq (rnhaque@gmail.com)
Abdul Jabbar (abdul.jabbar@aku.edu)
Tariq Moatter (tariq.moatter@aku.edu)

Version: 2 Date: 5 October 2010

Author's response to reviews:

1. Reformated abstract